

# THE UNIVERSITY of EDINBURGH

## Edinburgh Research Explorer

## Drug metabolism and pharmacokinetics of praziquantel

#### Citation for published version:

Zdesenko, G & Mutapi, F 2020, 'Drug metabolism and pharmacokinetics of praziquantel: A review of variable drug exposure during schistosomiasis treatment in human hosts and experimental models', *PLoS Neglected Tropical Diseases*. https://doi.org/10.1371/journal.pntd.0008649

#### **Digital Object Identifier (DOI):**

10.1371/journal.pntd.0008649

#### Link:

Link to publication record in Edinburgh Research Explorer

**Document Version:** Publisher's PDF, also known as Version of record

Published In: PLoS Neglected Tropical Diseases

#### **General rights**

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

#### Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





### G OPEN ACCESS

**Citation:** Zdesenko G, Mutapi F (2020) Drug metabolism and pharmacokinetics of praziquantel: A review of variable drug exposure during schistosomiasis treatment in human hosts and experimental models. PLoS Negl Trop Dis 14(9): e0008649. https://doi.org/10.1371/journal. pntd.0008649

**Editor:** John Stuart Gilleard, University of Calgary, CANADA

Published: September 25, 2020

**Copyright:** © 2020 Zdesenko, Mutapi. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This research is supported through the Precision Medicine Doctoral Training Program, GZ was funded by the Medical Research Council (MRC) (https://mrc.ukri.org/). The views expressed in this publication are those of the authors and not necessarily those of the MRC. This research was also commissioned in part by the National Institute for Health Research (NIHR) Global Health Research Program (16/136/33) using UK AID from the UK Government. The views expressed in this publication are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care. The funders had no role in REVIEW

## Drug metabolism and pharmacokinetics of praziquantel: A review of variable drug exposure during schistosomiasis treatment in human hosts and experimental models

#### Grace Zdesenko<sup>1,2</sup>\*, Francisca Mutapi<sup>1,2</sup>

1 Institute of Immunology & Infection Research, University of Edinburgh, Ashworth Laboratories, King's Buildings, Edinburgh, United Kingdom, 2 NIHR Global Health Research Unit Tackling Infections to Benefit Africa (TIBA), University of Edinburgh, Ashworth Laboratories, King's Buildings, Edinburgh, United Kingdom

\* g.zdesenko@sms.ed.ac.uk

## Abstract

Schistosomiasis control is heavily reliant on the drug praziguantel (PZQ), which is used as preventive chemotherapy as part of national helminth control strategies. Given the heavy reliance on PZQ for mass drug administration, there has been considerable research on the potential of parasites developing resistance to the drug, resulting in decreased drug efficacy. However, there have been comparatively fewer studies of other factors that can potentially alter PZQ efficacy. Here, we investigate whether host PZQ metabolism contributes towards variable cure rates. We evaluate factors that can influence the metabolism of PZQ and the resultant effect on the efficacy of PZQ treatment to determine factors that potentially influence an individual's response to the drug. The literature search was directed at published studies from three online databases: Web of Science, PubMed, and EMBASE. The search terms for the review comprised of ([praziguantel OR PZQ] AND [schistosom\* OR bilharzia] AND [pharmaco\*]) and included studies evaluating PZQ metabolism. Publications were categorised into pharmacokinetics, drug-drug interactions, pharmacogenetics, and metabolite analysis. Forty publications describing human and experimental studies fitted the inclusion criteria and were subjected to data extraction and analysis. The analyses showed that variable exposure to PZQ was associated with alterations in the liver's capacity to metabolise PZQ and observed drug-drug interactions. Other factors influencing the efficacy of PZQ were brand, formulation, and co-administered food. Although some work has been performed on metabolite identification, there was minimal information on PZQ's metabolic pathway, and no pharmacogenetics studies were identified. The study indicated that in both human and experimental studies alterations in the liver's capacity to metabolise PZQ as well as drug-drug interactions affected systemic levels of PZQ that could result in variable cure rates. The study confirmed previous findings of higher antischistosomal activity of (R)-PZQ enantiomer when administered alone compared to the racemate at the same dose as well as improved efficacy when the drug is administered with food. The study also highlighted

study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

the need for more comprehensive studies of the PZQ metabolic pathway and PZQ pharmacogenetic studies in humans.

#### Author summary

Schistosomiasis is a neglected tropical disease caused by parasitic worms, and its control is heavily reliant on the drug praziquantel (PZQ). Reports of individuals not being cured of schistosomiasis after a treatment have raised concerns about drug efficacy and the potential for the development of drug resistance. Many factors can be responsible for reduced efficacy including host, parasite, and drug-related factors. We investigated the potential effect of varying systemic levels of PZQ on cure rates. We reviewed studies on PZQ metabolism identifying factors that may influence PZQ systemic levels, including the drug administered, host liver health, and genetic polymorphisms of liver enzymes involved in PZQ metabolism. The results indicated that varying levels of PZQ as well as drug-drug interactions. The study also highlighted the need for further studies of the effect of human genetics on PZQ's metabolice pathway, and the PZQ metabolites produced.

#### Introduction

Schistosomiasis is a prominent public health problem [1], with the majority of affected people residing in Africa [2]. Praziquantel (PZQ) is the drug of choice to treat schistosomiasis and is widely used in preventive chemotherapy (PCT) programs (as defined by WHO) [3] across Africa to treat intestinal and urogenital schistosomiasis infections caused by *Schistosoma mansoni* and *S. haematobium* parasites, respectively [4]. Mass drug administration (MDA) of PZQ in PCT to treat schistosome infection and reduce associated morbidity has been a success, with an estimated 235 million people treated with PZQ in 2018 alone [5]. PZQ itself is a racemic drug, with the standard dose consisting of a 1:1 mixture of two enantiomers (see Fig 1).

Only the (R) enantiomer (also known as Levo-PZQ, L-PZQ, or (-)-PZQ) of PZQ has antischistosomal activity; in contrast, the (S) enantiomer (also known as Dextro-PZQ, D-PZQ or (+)-PZQ) does not have antischistosomal action but contributes to some of PZQ's known side effects [6]. PZQ is well-tolerated and effective in patients of all ages with different clinical forms of schistosomiasis [7] and has been used to treat schistosomiasis since the 1980s [8]. The precise mechanism of PZQ's antiparasitic action remains poorly described. Studies suggest that its action arises from the (R)-PZQ enantiomer disrupting the schistosome calcium ion homeostasis causing uncontrolled muscle contraction and death in adult worms [9, 10]. PZQ targets only adult schistosomes; therefore, the drug is not effective against the larval stages infections [11].

The efficacy of PZQ treatment is determined by the cure rate (CR), which compares the number of egg-positive individuals pre-PZQ treatment who become negative for schistosomiasis post-PZQ treatment as well as by the egg reduction rate (ERR), determined by the reduction in mean number of eggs excreted in urine or stool (depending on the schistosome species) from pre-PZQ to post-PZQ treatment [12, 13]. Low PZQ cure rates have been reported by some studies, with many attributing this to a high pretreatment parasite burden [14–17]. However, variable efficacy of PZQ has also been observed in other studies, even when accounting for the level of pretreatment infection [18, 19], suggesting that other factors must be influencing the drug's efficacy. For example, low cure rates and the reduced efficacy of PZQ



(R)-Praziquantel

(S)-Praziquantel

**Fig 1. The molecular structure of the two PZQ enantiomers.** The (R)-praziquantel has the hydrogen atom (H) pointing down from the chiral center. The (S)-praziquantel has the hydrogen atom (H) pointing up from the chiral center. PZQ, praziquantel.

https://doi.org/10.1371/journal.pntd.0008649.g001

treatment have been attributed to patients harbouring schistosomes in different developmental stages and the decreased sensitivity of schistosomes to PZQ treatment [20–23]. Investigations into cases of laboratory and field PZQ resistance indicated that reduced efficacy of drug treatment due to decreased sensitivity and resistance was rare [24, 25]; hence, parasite sensitivity cannot account for all incidences of treatment failure. So, why are these low cure rates occurring? This study investigates whether the level of PZQ in systemic circulation is contributing towards low cure rates, as systemic PZQ may not be exceeding the lethal schistosome concentration. In this study, we will focus on three factors that can influence the amount of PZQ in systemic circulation, and, thus, affect the efficacy of PZQ treatment: PZQ pharmacokinetics (PK), pharmacogenetics, and drug–drug interactions.

The term "PK" describes movement of a drug, encompassing the absorption, distribution, metabolism, and elimination (ADME) parameters of that drug [26]. Metabolism describes the mechanism of breaking down the drug compound and is commonly analysed via biotransformations of the xenobiotic by drug-metabolising enzymes [27]. The Cytochrome P450 (CYP) enzymes mediate the metabolism of PZQ, specifically the following enzymes: CYP1A2, CYP2C9, CYP2C19, CYP2D6, CYP3A4, and CYP3A5 [28]. The term pharmacogenetics refers to the variability in an individual's response to a drug due to genetic variations. Genetic polymorphisms in CYPs have linked interindividual variation to metabolism in numerous drug efficacy and toxicity studies [29]. These variations are primarily due to single nucleotide polymorphisms (SNPs) of the CYPs, and these can lead to an increased or decreased pharmacokinetic effect [30]. The distribution of CYP alleles differs substantially between populations, emphasising the need for optimising drugs for the population in which the drug will be used, e.g. the efficacy of drugs tested in Europe may not have the same efficacy in African populations [31]. Furthermore, analysing pharmacogenetic differences in the metabolic (drug metabolism) products can provide additional information on a patient's drug response [32].

The CYPs are also an important site for drug–drug interactions (DDIs) [33]. PZQ is metabolized by multiple CYPs, and so DDIs within these CYP pathways could result in the formation and accumulation of metabolic by-products or a reduction of the drug's therapeutic effect [34]. The bioavailability of a drug can be altered by DDIs via competition for protein-binding sites on the CYPs, affecting the overall efficacy of a drug. DDIs can induce the CYPs (increased metabolism), increasing the activity of the CYP enzymes and decreasing the overall bioavailability of the active drug [35]. Vice versa, if a coadministered drug inhibits the PZQ's enzymatic binding sites, it can no longer be metabolised and eliminated via this pathway (decreased metabolism), and drug accumulation could result in toxic levels in the body [36]. Evaluating DDIs is important when developing and using any drug treatment, especially when regarding comorbidity, as patients may be on multiple medications [37]. Overall, evaluating whether pharmacogenetics factors and DDIs influence a CYP's ability to metabolise PZQ will indicate the role of altered metabolism in variable schistosomiasis CR.

#### Methods

#### Literature search strategy

The search strategy was guided by the Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) guidelines [38], and the studies included in this review were published prior to May 12, 2019. The search used three online databases: Web of Science, PubMed, and EMBASE. The search terms for the review comprised of ([praziquantel OR PZQ] AND [schistosom\* OR bilharzia] AND [pharmaco\*]) and included animal and experimental models that evaluated PZQ metabolism. As the main burden of schistosomiasis in sub-Saharan Africa is attributed to two species of schistosomes [4], only studies on *S. haematobium* and *S. mansoni* were included in this review. The citations were compiled in EndNote X8 and duplications were identified and removed.

#### Literature screening and inclusion and exclusion criteria

The titles and abstracts of the Endnote Library were screened, and any that met the inclusion criteria were put aside for a review of the full text. If the abstract was not available or it was unclear whether the study fully met the inclusion criteria, it was selected for full evaluation. After a full evaluation of each text, the articles included in this review were required to fit into one of the following topics: PK, pharmacogenetics, drug–drug interactions, or metabolites analysis. Selection criteria within each category were based on whether (1) the article was available as full journal text and in English, (2) the units of the numerical pharmacokinetic parameters were clearly defined, (3) the article included human and animal *in vivo/in vitro* models for human extrapolation, and (4) models were either a healthy control or infected with *S. mansoni* or *S. haematobium*. If a paper could not be located through an online repository, it was requested from the British Library. Any papers that could not be obtained from these sources were excluded.

#### Data extraction and data analysis

Pharmacokinetic data, including the dose of PZQ and any "Drug B" measured (racemic or enantiomeric PZQ), the model, *Schistosoma* infection status, the number of subjects (N), the numerical pharmacokinetic parameter and its units, and any external test conditions, such as fasting or fed states, were documented in an Excel spreadsheet. To extract comparable data from each study, it was concluded that the *in vivo* pharmacokinetic parameters to quantify drug-concentration-time relationships would be as follows: area under the plasma-concentration curve (AUC), peak plasma concentration ( $C_{max}$ ), time to reach peak plasma concentration following drug administration ( $T_{max}$ ), and drug-elimination half-life ( $t_{1/2}$ ) [39]. Data points were extracted, grouped together by parameter, and converted into standard SI units. If no units were quoted, the data point was excluded from the data tables. Where the published data was presented in a graph, Data Thief III software was used to extract the numerical values. The data points were analysed to detect trends affecting PZQ metabolism between studies, creating new combined graphical results. These graphs were then analysed in the results to assess

whether the combined data reflected the hypothesis of this review. Only limited statistical analysis was possible due to sparsity of data and heterogeneity of populations used.

To evaluate the efficacy of PZQ without pharmacokinetic values, the worm burden reduction (WBR) in the case of experimental models, ERR, and CR were also collected. For pharmacogenetic factors, the population ethnicity of each human study was recorded when provided in the source publication as genetic factors can determine response to a drug [40]. By linking the individuals in the studies by ethnicity, it implies a common genetic ancestral background that could influence a therapeutic response to a drug [41]. For the papers that depicted PZQ metabolite analysis, the structural identity of each metabolite was created using ChemDraw Prime v16.0, and any information about its metabolic pathway was reported.

#### Results

The search yielded a combined total of 873 studies and, after the removal of duplicates, 425 titles and abstracts were screened, resulting in 95 publications for a full-text review. There were 55 excluded publications, leaving 40 publications for data extraction and analysis (S1 Table). The results of this systematic search are displayed in a flow chart (S1 Fig).

#### PK and PZQ efficacy

In the PK category of this review, data points extracted from multiple papers describing healthy and infected human and animal models provided an overview of the published PK and efficacy of PZQ treatment. The animal models obtained from the search strategy were exclusively rodents, and included mice, rats, and hamsters. The human studies included adults (16 years or older), school-aged children (SAC) (6 to 15 years), and preschool-aged children (PSAC) (5 years or younger). An evaluation of these as separate study groups was made based on the model, either rodent or human, and parameters measured.

#### **Experimental studies: PK and PZQ efficacy**

The relative exposure of PZQ in the reviewed rodent models was compared using  $t_{1/2}$ ,  $T_{max}$ ,  $C_{max}$ , and AUC values (refer to S2 Table). The effects of different PZQ doses, brands, and infection status on the exposure parameters ( $C_{max}$  and AUC) of PZQ in various rodent models were described in six studies [42–47].

Hepatic CYP metabolism. On average, based on the PK data extracted from these papers, the general trend of the compiled exposure parameters was that with increasing PZQ dose, the C<sub>max</sub> and AUC also increased. Upon closer inspection, there were some deviations from this trend with a common factor; it appeared that the healthy rodent models tended to have a lower exposure to PZQ than the infected rodents. A study by Botros and colleagues [42] in healthy and S. mansoni-infected mice showed that in the infected mice there was a significant difference in maximum concentration of PZQ in systemic circulation, approximately triple that found in healthy mice. The increased exposure was attributed to the reduced liver capacity of infected rodents, resulting in decreased metabolism of parent PZQ and, therefore, longer exposure in the circulation. Botros and colleagues [43] and Kokwaro and colleagues [44] conducted further analysis of the overall decreased hepatic CYP activity and the resultant alteration of PZQ's PK parameters due to schistosomiasis, once again with significant differences between S. mansoni-infected and uninfected mice detected. The relative t<sub>1/2</sub>, T<sub>max</sub>, C<sub>max</sub>, and AUC values were increased proportionally to the degree of decreased CYP activity, an influential factor if this effect is also observed in humans. In fact, due to decreased hepatic CYP activity, severe side effects were observed by Gotardo and colleagues [45] from toxicity caused by the higher  $C_{max}$  and AUC values in infected mice treated with 400 mg/kg PZQ.

**PZQ brands.** Additional factors which may have contributed to variable PZQ exposure were investigated by comparing the parasitic efficacy of the different brands of PZQ available. To compare efficacy, two parameters were measured: (1) WBR, as the World Health Organization describes a reduction in adult worm counts as a measure of the efficacy of an anthelmintic in experimental models [48], and (2) the percentage inhibition of hepatic CYP450 activities as a result of *S. mansoni* infection, observing the extent of CYP450 inhibition when compared to healthy mice. The results in S3 Table are adapted from Botros and colleagues [42], with the highest WBR and lowest CYP450 inhibition was achieved with pure PZQ, with Distocide and Biltricide showing comparable efficacy. Interestingly, Bilharzid, brand T3A, and Epiquantel had a significantly lower WBR than the other brands analysed, yet had significantly higher CYP450 inhibition. In comparison to pure PZQ, Distocide and Biltricide, the Bilharzid, brand T3A, and Epiquantel were not aiding the recovery of the CYP450 activity from schistosomiasis infection as well as the other brands examined.

**PZQ formulations.** Further studies focused on the PZQ compound itself as a topic of discussion, with Zhang and colleagues [46] and Meister and colleagues [49] showing the pharma-cological differences between the (R) and (S)-PZQ enantiomers, confirming the antischistosomal activity of (R)-PZQ. A comparison of the activity of racemic PZQ (rac-PZQ) compared to each PZQ enantiomer in mouse models can be seen in Fig 2 (data in S4 Table).

This is consistent with (R)-PZQ being the pharmacologically active enantiomer, with a significantly higher WBR than the (S)-PZQ and a higher WBR than the current standard rac-PZQ treatment at the same dose [49, 50]. One contradictory piece of data extracted during this review was from Tanaka and colleagues [50], in which the (S)-PZQ had a higher WBR than rac-PZQ and (R)-PZQ at 50 mg/kg of PZQ, yet this could be due to interanimal variability as the study was only conducted in seven mice. Overall, this discrepancy in the activity of the enantiomers was only observed in one paper and was not seen in any of the human studies.

A PZQ formulation (Polymorph B) was tested in mice by Lombardo and colleagues [47] as a new treatment option in comparison to the current PZQ tablet. Polymorph B aimed to increase efficacy and improve the PK parameters of the current PZQ formulation, which has low bioavailability and low water solubility, by enhancing PZQ's solubility and dissolution. The crystalline formulation created from grinding rac-PZQ showed improved physical properties, particularly increasing PZQ chemical stability and doubling water solubility. Yet, the PK parameters of both enantiomers of Polymorph B had a lower exposure profile compared to a reference PZQ, indicating lower bioavailability of the new drug formulation compared to the current standard PZQ. This was most prominently visualised in the AUC value, which was approximately 40% lower for Polymorph B than reference PZQ.

#### Human studies: PK and PZQ efficacy

As with the rodent models, the relative exposure of PZQ was compared using  $t_{1/2}$ ,  $T_{max}$ ,  $C_{max}$ , and AUC values, with 13 papers containing human PK data sets [51–63].

**PZQ formulations.** As seen in the rodent models, data in human studies also showed a significant difference in antischistosomal activity between the two PZQ enantiomers. Four papers [51–54] (S5 Table) measured the PK of each PZQ enantiomer after a racemic dose in humans. The results were compiled and, as Fig 3 depicts, there were clear differences in enantiomer exposure between the biologically active (R)-PZQ and the (S)-PZQ.

Despite (R)-PZQ being the desired circulating pharmacoactive substance, the results extracted during this review in healthy normal volunteers (HNV) showed that the exposure parameters of the (R)-PZQ enantiomer were significantly lower than the (S)-PZQ. Therefore, the C<sub>max</sub> and the AUC of the active enantiomer are lower than the inactive enantiomer at the



Fig 2. Comparison of the percentage WBR in an infected mouse model when dosing racemic PZQ or a single enantiomer. The data for this graph were extracted from two papers in this review and evaluated to assess the WBR-dose response [49, 50]. The relationship between WBR and drug formulation was found to be significantly different (\*P < 0.01, \*\*P < 0.001) for each PZQ dose based on the results of the independent samples Kruskal-Wallis Test. PZQ, praziquantel; WBR, worm burden reduction.

same dose (Fig 3), so, when PZQ is dosed as a racemic tablet, there are significantly lower proportions of the desired enantiomer entering the circulation. This is the opposite of the desired action and demonstrates that a dose with a rac-PZQ tablet is not reflective of the dose of active (R)-PZQ.

Two studies in this review evaluated the PK of a small orally dispersible tablet (ODT) formulation that would allow for fast dispersion and an acceptable taste [51, 52]. The relative bioavailability of a rac-PZQ ODT formulation was compared with the single active enantiomer (R)-PZQ ODT ((R)-ODT), to determine if there were pharmacokinetic differences between ODTs and the current rac-PZQ tablet. When dosed at 40 mg/kg, the rac-PZQ reference is 50:50 of (R)-PZQ to (S)-PZQ; therefore, it was expected that when the (R)-ODT was dosed at 20 mg/kg, it would be approximately equal to that of the 40 mg/kg of rac-PZQ or the PZQ



Fig 3. The mean AUC values extracted from the included studies against the PZQ dose of each of the PZQ enantiomers in HNV. The PZQ enantiomers were measured after a racemic PZQ dose or the (R)-PZQ dosed alone. These data were extracted from multiple papers in this review, separated by PZQ enantiomer, and the AUC was averaged for each dose and plotted on a graph for analysis [51–54]. Using a linear regression model, the difference in AUC between (R)-PZQ and (S)-PZQ in humans was found to be statistically significant (P < 0.05). AUC, area under the curve; HNV, healthy normal volunteers; PZQ, praziquantel.

reference. However, this did not appear to be the case as the 20 mg/kg dose of (R)-ODT was only around 40% of the PZQ reference.

**Hepatic CYP metabolism.** The activity of the drug-metabolising CYPs during PZQ metabolism and the resultant variable exposure of the drug was assessed by multiple studies. With regard to hepatic CYP activity, el Guiniady and colleagues [55] investigated the potential decrease in CYP metabolism in *S. mansoni*–infected adult patients with either liver cirrhosis or splenomegaly (data in S6 Table), building on the rodent study by Kokwaro and colleagues [44]. The study concluded that, in general, the  $C_{max}$ ,  $T_{max}$ , and AUC were significantly higher in the liver-impaired patients compared to the controls. This was determined to be due to the delay in elimination of PZQ to its metabolites as a result of decreased CYP function [55]. In humans, as the severity of cirrhosis increased, the patients had an elevated  $C_{max}$  and AUC in comparison to HNV and nonimpaired *S. mansoni* patients at the same PZQ dose.

Concerning the hepatic metabolism of PZQ in children, Kovac and colleagues [56] investigated the PK parameters of SAC and PSAC regarding age-related metabolism and the level of CYP maturity in children. Data points from three papers [56–58] were obtained (S7 Table), including the  $t_{1/2}$ ,  $T_{max}$ ,  $C_{max}$ , and AUC. The increase in AUC with increased dose for SAC and PSAC infected with *S. mansoni* and *S. haematobium* were compared in Fig 4, measuring



Fig 4. The mean AUC values extracted from the included studies against PZQ dose of (R)-PZQ [I, II], (S)-PZQ [III, IV], and the major metabolite (R)-trans-4-OH-PZQ [V, VI] at 20, 40, and 60 mg/kg. The data are extracted from Kovac and colleagues [56], which investigated *S. mansoni* and *S. haematobium* infected PSAC ( $\bullet$ ) and SAC ( $\circ$ ) treated with PZQ. The AUC values were then plotted against the dose of PZQ administered for further analysis. *P* < 0.05 was considered statistically significant; a: significant difference between SAC and PSAC for the same dose and analyte, b: significant difference between *S. haematobium* and *S. mansoni* for the same dose, age group, and analyte, and c: significant difference between (R)-PZQ and (S)-PZQ for the same dose, age group, and species. AUC, area under the curve; PSAC, preschool-aged children; PZQ, praziquantel; SAC, school-aged children.

the (R)-PZQ (Fig 4[I, II]), (S)-PZQ (Fig 4[II, IV]), and the major metabolite (*R*)-4-OH-PZQ (Fig 4[V, VI]).

There was a significant difference between the AUC values of the infected *S. mansoni* SAC and PSAC at the same dose and analyte, with the exception of (S)-PZQ at 40 mg/kg. Additionally, the AUC of (R)-PZQ and (S)-PZQ in *S. mansoni* SAC was found to be significantly different at the same dose but only at 60 mg/kg for PSAC. Generally, the AUC values of the infected *S. mansoni* PSAC were higher than the infected *S. mansoni* SAC for both the PZQ enantiomers (Fig 4[I, III]) [56], suggesting that once dosed with PZQ, PSAC metabolised the drug more slowly. This is in accordance with the age-related model by Bonate and colleagues [52], which stated that due to the decreased CYP activity of the PSAC, the drug remains in circulation longer, resulting in a higher AUC. Conversely, the AUC for the (*R*)-4-OH-PZQ metabolite was higher in SAC because they had potentially metabolised the drug more rapidly than PSAC (due to increased CYP activity), leading to a higher circulating level of the metabolite (Fig 4 [V]). Regarding the exposure in *S. haematobium*-infected children, there was no significant difference between exposure levels in *S. haematobium*-infected PSAC and SAC (Fig 4[II, IV, VI]).

**PZQ dosing.** In infected children, PZQ is delivered at a dose extrapolated from adult tolerance studies. Bustinduy and colleagues [58] and Bonate and colleagues [52] both described the limitations of current treatment models and further investigated whether the dosing regime was contributing to low CR in children (S8 Table). Both studies highlighted that the extrapolation of infected African children from healthy European adult volunteers is misjudged, and the resultant model cannot predict the differing bioavailability in each population. The resultant extrapolation model concluded that the standard method was not predictive of treatment success and that further studies are required to optimise PZQ treatment, looking at the effect of higher dosage, sex, weight, and PZQ enantiomeric activity [58]. Additionally, Bonate and colleagues [52] described the potential of using a model that includes the degree of maturation of the CYP isoforms involved in PZQ metabolism, and aimed to use this model to predict PZQ exposure (AUC) based on age.

**Impact of food.** There are additional factors which need to be considered that could affect PZQ efficacy, specifically factors not attributed to the host biology. The varying fasting and fed state of multiple studies were found to contribute to variable exposure of PZQ, noting that the bioavailability of PZQ increases with the administration of food [64]. Mandour and colleagues [63] confirmed that, after oral administration in HNV, the bioavailability is reliant on food intake, with PZQ clearance affected by content of the diet. A high oil diet enhanced PZQ absorption, and a high carbohydrate diet appeared to inhibit CYP activities due to an accumulation of metabolites, preventing further metabolism and allowing PZQ to remain in systemic circulation longer.

#### **Drug-drug interactions**

There is a paucity of information on DDIs with PZQ, with only 17 published papers on the topic. Of these studies, many evaluated the use of DDIs to lower PZQ dosing regimens for a combined drug treatment. To define the effect of each drug combination, an effect-based strategy was introduced, which followed the principle that if one component alters the ability of PZQ to reach the necessary lethal schistosome concentration via alterations to PZQ metabolism, then this action was designated a pharmacological effect [65]. Therefore, this review used a pharmacokinetic-based assessment of DDIs based on the alteration of CYP activities and potentially the overall efficacy of PZQ treatment [66]. This "bioavailability model" focused on the change in the AUC due to these drug combination effects to aid further discussion of



**Fig 5. The DDIs identified by the review, and the resultant effect on PZQ exposure.** The *in vivo* model and "Drug B" are listed as well as the overall effect of the drug combination on exposure of PZQ. The effect of the drug combination on PZQ efficacy in rodents was calculated based on the percentage change in WBR of each drug alone in comparison to the combined action (S9 Table) [73–83]. The effect of the drug combination on PZQ efficacy in humans was calculated based on the percentage change in AUC of each drug alone in comparison to the combined action (S10 Table) [54, 59, 84–86]. AUC, area under the curve; DDI, drug–drug interaction; PZQ, praziquantel; WBR, worm burden reduction.

potential DDIs that could alter PZQ efficacy [67, 68]. In combination with the descriptive results of each DDI by the papers reviewed, this allowed a comparison of the potential combined effect of the two drugs ( $E_{PZQ+B}$ ) to the effects of its individual components ( $E_{PZQ}$  and  $E_B$ ) [69]. The effect of the DDI is expressed as (1) a synergistic effect: the combined therapy has a greater therapeutic effect than each drug alone following the principle  $E_{PZQ}+E_B\ll E_{PZQ+B}$ , where  $E_{PZQ+B} > 20\%$  change in AUC compared with PZQ alone; (2) an additive effect: similar to each drug used individually, no significant increase in activity  $E_{PZQ}+E_B\approx E_{PZQ+B}$ ; (3) an antagonistic effect: interactions that could decrease therapeutic efficacy, in which  $E_{PZQ+B}<-20\%$  change in AUC compared with PZQ alone, or (4) no effect to treatment, these results were not outside the threshold of  $-20\% \le E_{PZQ+B} \ge 20\%$  change in AUC compared with PZQ alone [70–72]. The PZQ drug combinations obtained in this review are listed in Fig 5, depicting the result of the drug on PZQ action as described by the published DDI studies.

#### Experimental studies: drug-drug interactions

**Synergistic and additive drug-drug interactions.** To reduce the dose of PZQ while still having an efficacious effect, synergistic activity using the antimalarial mefloquine was explored in rodents by El-Lakkany and colleagues [73]. The WBR of PZQ + mefloquine was over double the value of PZQ alone, indicating a synergistic effect that maintained lethal schistosome PZQ concentrations and that could be applicable to humans treated with both drugs. Furthermore, when PZQ combined with another antimalarial, artemether, synergistic and additive effects were observed in separate rodent studies, even when PZQ was reduced to around a quarter of its recommended dose. A PZQ + artemether study by Utzinger and colleagues [74] concluded that the lower doses used in the combined treatment was safer and more efficacious (with over double the WBR) than PZQ alone and was suggested as a basis for a human clinical trial [74, 75].

Another study aimed to decrease the dose of PZQ, with Keiser and colleagues [76] combining nilutamide with PZQ to increase WBR compared to PZQ alone. It appeared that at low doses the combination had an additive effect on PZQ, with nilutamide obtaining a higher WBR alone. Interestingly, a synergistic effect was seen with PZQ + nilutamide

(100 mg/kg + 200 mg/kg) dose, with a 67% increase in the WBR than PZQ alone at that same dose, something that was not observed for nilutamide below 200 mg/kg.

Further additive effects were evaluated by El-Lakkany and colleagues [77] and Botros and colleagues [78], using pentoxifylline and adamantylamide dipeptide in combination with PZQ (S11 Table). Both combinations used a subcurative dose of PZQ in the DDI to evaluate the enhancement of each drug on PZQ's therapeutic effect, with both combinations showing comparable results to the full dose of PZQ [77, 78]. Abla and colleagues [79] co-dosed PZQ with 1-aminobenzotriazole, a pan-CYP inhibitor, and predicted a reduction in PZQ clearance and in turn increase  $C_{max}$  and AUC of the pharmacoactive (R)-PZQ in plasma. Yet, in *S. mansoni*-infected mice, 1-aminobenzotriazole was found to have only a small impact on the WBR, only increasing it by approximately 25%, reaching similar levels to PZQ alone (additive) taking into account interanimal variability. This was similar to the combination of turmeric and PZQ [80], in which a significant additive effect was observed, with around a 24% increase in WBR.

Antagonistic drug-drug interactions. Masimirembwa [81] and colleagues reported two antagonistic effects with PZQ in rats, as combinations with both phenobarbital and 3-methyl-cholanthrene decreased the  $C_{max}$  and AUC of PZQ, with phenobarbital decreasing PZQ exposure greater than 3-methylcholanthrene.

**No effect.** It is also important to analyse the combinations that have no effect on PZQ efficacy and can, therefore, be confidently co-administered without affecting PZQ's systemic concentration. Araujo and colleagues [82] and Ebeid and colleagues [83] investigated the combination of PZQ with clonazepam and metrifonate. Both studies concluded that there was no beneficial synergistic or additive action with the combined treatments of PZQ and that any antischistosomal action originated from PZQ alone and not a DDI. Furthermore, dexamethasone, a multiple CYP inducer, was expected to antagonistically decrease PZQ exposure and decrease PZQ efficacy due to an increase in the CYPs available to metabolise PZQ. Contrary to these predictions by Abla and colleagues [79], dexamethasone decreased plasma exposure of (R)-PZQ by approximately 10 fold but did not affect overall PZQ efficacy in rodents.

#### Human studies: drug-drug interactions

Five studies [54, 59, 84–86] describing human DDIs and the resultant variable drug exposure after oral administration of drug and food combinations were examined. The drug combinations and the percentage change in AUC from PZQ are shown in Fig.6.

**Synergistic DDIs.** Nleya and colleagues [84] evaluated the effect of combining PZQ with ketoconazole, as ketoconazole is a prominent inhibitor of the CYP3A4/5 isoforms which are known metabolic pathways of PZQ [88]. There was a 75% increase in the relative bioavailability of PZQ (measured by the change in the AUC) when co-dosed with ketoconazole, with the  $C_{max}$  increasing by 96% in response to the DDI. To measure the extent of the synergistic effect, the study was dosed at one-half the recommended value for a PZQ treatment (20 mg/kg PZQ + 200mgs ketoconazole). Despite this, 9 of 29 individuals reached the 1µM therapeutic threshold required to kill adult schistosomes, compared to only 2 of 29 when PZQ was given alone.

PZQ + cimetidine were evaluated by Jung and colleagues [59], with the potential to simultaneously treat schistosomiasis and neurocysticercosis, building on a study by Metwally and colleagues [85] that noted elevated PZQ sera concentrations with this DDI. Compared to PZQ alone, the plasma levels of PZQ more than doubled during combined drug administration, with a  $C_{max}$  greater than 400 ng/ml even after 12 hours, suggesting a synergistic effect.

Lima and colleagues [54] treated HNV with a combination of PZQ and albendazole (anthelmintic used for treating soil-transmitted helminths), with the aim of improving the therapeutic efficacy of both drugs by increasing plasma concentrations of the active forms of both drugs.



**Fig 6.** A summary of the human DDIs. Each bar represents the percentage change in AUC during the drug combination in comparison to PZQ alone. The effect of the drug combination is listed above the bar chart, with ±20% representing the boundaries of no effect due to interindividual variation [87]. The data were extracted from five papers [54, 59, 84–86], compiled in S10 Table, analysed to create percentage change in AUC and then plotted to aid further analysis. ABZ, albendazole; AUC, area under the curve; BIC, bicarbonate; CHQ, chloroquine; CIM, cimetidine; DDI, drug–drug interaction; GLUC, glucose; KTZ, ketoconazole; PZQ, praziquantel.

This conclusion can be visualised in Fig 6, in which the combination of PZQ + albendazole appears to have no effect on the AUC of rac-PZQ. Opposing this, Lima and colleagues [54] showed that PZQ increased the  $C_{max}$  of the active albendazole metabolite, and, in turn, albendazole increased the AUC of the pharmacologically active (R)-PZQ: a synergistic effect. In Fig 6, the AUC of (R)-PZQ increased by 76%, which would not be apparent if only the rac-PZQ was measured.

Antagonistic drug-drug interactions. Masimirembwa and colleagues [86] investigated the antagonistic effect of the CYP inhibitor chloroquine in HNV, with chloroquine expected to increase the bioavailability of PZQ due to its action as a CYP3A4 inhibitor. In contrast, it was found to decrease the  $C_{max}$  and AUC of PZQ, decreasing overall exposure, with 50% of individuals studied not reaching the lethal schistosome plasma concentration of 1µM. Masimirembwa and colleagues [86] determined this effect was not due to alterations in CYP activities but, rather, an alternate mechanism of chloroquine.

An antagonistic effect was also observed by Metwally and colleagues [85], when PZQ was co-dosed with glucose. However, as depicted in Fig 6, the percentage change of the AUC for

the combination compared to PZQ alone was -27%, which was just outside of the 20% "no effect" threshold [87]. Therefore, although this has been noted as an antagonistic effect with glucose, it may be due to interindividual variation slightly skewing the results because there were only 20 participants in this study, and there may be no effect regarding this interaction.

**No effect.** Although the combination of bicarbonate with PZQ did reduce the AUC of PZQ [85], it was not outside the 20% threshold (see Fig 6). Similar to glucose, this combination is not likely to have an effect on the efficacy of PZQ and is most likely due to the mean calculation of the AUC and the interindividual variation in the metabolism of the participants in the study.

#### Pharmacogenetics

**Human studies: pharmacogenetics.** The search did not yield any studies targeting the pharmacogenetics of PZQ metabolism. However, the data extracted from the included studies contained numerous examples of metabolic variability in the human studies. Although there were no specific studies on pharmacogenetics, the reasoning behind the variable drug exposure and efficacy of PZQ was postulated by many authors to be attributed to interindividual variation and host genetic factors [43, 45, 55–59, 61–63, 75, 85, 86]. The majority of the included studies did not evaluate this hypothesis further but merely stated that CYP polymorphisms and CYP-related maturity may be causing variation in PZQ PK parameters and bioavailability. The ethnicities of the individuals included in the human studies in this review were also recorded. Of the 17 human studies, 17.6% studied individuals from Europe, 17.6% from South America, and 64.7% from Africa. The ethnicities of the study populations included in this review were recorded to aid further discussion.

#### Metabolite analysis of praziquantel

Seventeen metabolites were elucidated from four papers [28, 53, 84, 89], and their structures are displayed in Fig 7. PZQ and its enantiomeric metabolites, including their CYP pathway and structure, were evaluated by multiple *in vivo* and *in vitro* techniques. Using human plasma, human urine, human and mouse liver microsomes, and human recombinant enzymes, there were 8 distinguishable mono-oxidised metabolites, 2 dehydrogenated mono-oxidised metabolites, 3 di-oxidised metabolites, and 4 glucuronide metabolites detected.

A study by Nleya and colleagues [84] depicted the metabolic pathway of PZQ to its main metabolite; 4-OH-PZQ (Fig 7[II]). This study showed that when PZQ was concomitantly administered with CYP3A4/5 inhibitor ketoconazole, significant changes in the level of metabolites in circulation were observed. The AUC of the main 4-OH-PZQ metabolites were increased by 57% (*cis-*) and 67% (*trans-*) when co-dosed with ketoconazole, with 30 times more *trans* isomer than the *cis* as seen in Fig 8. The higher levels of the *trans-* isomer stems from the favourable cyclohexane ring conformation placing the bulky OH substituent equatorial to prevent steric hindrance compared to the axial (*cis-*) conformation [90].

#### **Discussion and conclusion**

Heterogeneity in PZQ drug efficacy has been reported in multiple studies [18, 19]. Given the concern about the development of drug resistance in schistosome parasites, there has been considerable research on the potential for parasites developing resistance. However, it is critical to investigate the impact of nonparasite related factors that give rise to variability in PZQ drug efficacy. The majority of studies determined this to be due to interindividual variation and host genetic factors with no further analysis [43, 45, 55–59, 61–63, 75, 85, 86]. Here, we determined whether variable cure rates in humans could be attributed to fluctuating levels of



**Fig 7.** A comprehensive map of PZQ metabolites extracted and combined from studies that focused on the metabolite analysis of PZQ. I: 8-OH-PZQ [89], II: 4-OH-PZQ [28, 53], III: X-OH-PZQ [84], IV: O-PZQ [28], V:O-PZQ [28], VI: O-PZQ [28], VII: O-PZQ [28], VII: O-PZQ [28], IX: (-2H)-O-PZQ [28], X: (-2H)-O-PZQ [28], X: (-2H)-O-PZQ [28], XIII: O\_2-PZQ [28], XIII: O\_2-PZQ [28], G1: Gluc-PZQ [28], G2/G3: Gluc-O-PZQ [28], G4: (-2H)-Gluc-O-PZQ [28]. The chemical structures and metabolite map were created using ChemDraw Prime v16.0. Gluc, glucuronide; PZQ, praziquantel.



Fig 8. The favourable formation of *trans*-4-OH-PZQ compared to the *cis*-4-OH-PZQ isomer created from pharmacokinetic data by Nleya and colleagues [84]. The equatorial position points upwards and the axial position points downwards. The chemical structures were created using ChemDraw Prime v16.0. PZQ, praziquantel.

https://doi.org/10.1371/journal.pntd.0008649.g008

PZQ in systemic circulation and explored the factors that influence an individual's response to PZQ treatment, focusing on the metabolism of the drug and the resultant efficacy of treatment.

Multiple studies reviewed confirmed (R)-PZQ as the pharmacologically active enantiomer and that the exposure of the desired (R)-PZQ enantiomer was significantly lower than the (S)-PZQ when dosed as rac-PZQ [46, 49-54]. The reasoning behind the chiral differences in the elimination of each enantiomer has been postulated to be due the variations in their affinity in binding to the active site of the CYPs [28]. If the affinity of (R)-PZQ for CYP1A2 is higher than (S)-PZQ, then the metabolism of the active (R)-PZQ to its major metabolite (as described by Nleya and colleagues [84]) would be greater than metabolism of (S)-PZQ, resulting in a more rapid decrease in (R)-PZQ plasma exposure than for (S)-PZQ. However, as affinity to CYP1A2 has not yet been evaluated, the metabolic differences between (R)-PZQ and (S)-PZQ are still unclear. These studies highlight the benefits of treatment with the (R)-PZQ enantiomer alone in comparison to using the racemate and supporting the development of new formulations with the active enantiomer, as is the case for the ODT paediatric formulations of PZQ currently undergoing clinical trials [91]. In this case, the ODT formulation increased the AUC of the desired (R) enantiomer for the rac-ODT and (R)-ODT in comparison to the reference rac-PZQ, but it had a nonproportional PK profile [51, 52]. This limited the interpretation of each PZQ enantiomers PK profile in the ODT and emphasises the need for an (R)-ODT dosefinding study in PSAC to fully define the PK in paediatric patients and find the correct dosing regimen.

In experimental models, host schistosome infection status was an important factor influencing PZQ's pharmacokinetic parameters, with PZQ's exposure parameters in infected animals higher on average than in uninfected animals [42-45]. The studies proposed that *S. mansoni* infection interferes with liver function, preventing PZQ metabolism. However, this does not easily translate to human hosts. Instead variable exposure in human hosts tended to be associated with alterations in the liver's capacity to metabolise PZQ due to other factors, as observed in studies in which liver disease impaired CYP function, leading to increased levels of PZQ in systemic circulation. In patients suffering from liver cirrhosis and also treated with PZQ, the C<sub>max</sub> and AUC increased with the severity of cirrhosis in comparison to nonimpaired *S. mansoni*–infected and healthy patients at the same PZQ dose [55]. The increase in exposure parameters is contrary to the study performed in rats by Kokwaro and colleagues [44] and further highlights the limitations of experimental models in capturing human dynamics.

In human studies, age-related maturation of hepatic CYPs has been postulated, suggesting that the hepatic CYP pathways are not fully matured in children and affect drug AUC [52, 58]. The effect of age-related metabolism was investigated in a study of *S. mansoni*–infected children in which the AUC values of the PSAC were higher than the SAC for both the PZQ enantiomers in the circulation, yet the AUC for the (*R*)-4-OH-PZQ metabolite was higher in SAC [56]. This suggested PSAC could be metabolising the drug more slowly than the SAC, with higher concentrations of the parent drug found in PSAC and higher concentrations of metabolite found in SAC. This is consistent with higher exposure to active PZQ in the PSAC, resulting in higher cure rates compared to SAC. This is contrary to a study performed by Coulibaly and colleagues in which the cure rates in PSAC were lower at 40 and 60 mg/kg than in the SAC. This concurred with the data from this review at 60 mg/kg, as the SAC had a 5% higher cure rate than PSAC. However, Coulibaly and colleagues did suggest that during the study the crushing of the tablets may have altered the PK of PZQ, which may account for the difference in efficacy [92]. This age-related effect has not yet been observed in *S. haematobium*. Ofori-Adjei and colleagues [57] reported no differences in the PK parameters between uninfected

and *S. haematobium* infected SAC, concluding that *S. haematobium* infection does not influence PZQ metabolism due to alterations to hepatic function [56].

Non-host-related factors also affected exposure and efficacy of PZQ treatment, and these included PZQ brands and the coadministration of food or other drugs. Simultaneous administration of PZQ and food can alter the PK as food can delay absorption, affect stomach pH, alter blood flow, or interact with the PZQ itself [93]. One of the initial PZQ tolerance studies in humans provided the adult volunteers with a standardised meal for continuity in the study [61], but during PCT programs, this is not particularly representative, as diet can vastly differ. Fed groups have a lower clearance of PZQ due to altered absorption, highlighting the importance of attaining a fed state during treatment and in human PK studies to enhance success of exposure to the anthelmintic [63]. Of the different brands of PZQ on the market, Distocide and Biltricide were found to be more comparable in terms of exposure to pure PZQ [42]. Even so, the PZQ brands evaluated showed a decreased AUC,  $C_{max}$ , and  $t_{1/2}$  and, therefore, decreased efficacy and drug exposure in comparison to pure PZQ, indicating that treatment failures may be due to variable exposure of PZQ arising from the quality of the PZQ formulation.

A host attribute that can also affect drug metabolism is genetic polymorphism of liver enzymes involved in drug metabolism, i.e., pharmacogenetics. While CYP pharmacogenetic studies have been conducted in different ethnic populations [94], there are relatively few studies on PZQ metabolism. [94]. In recent publications, trials of more efficacious formulations and age-dosing models were being performed in European populations instead of the majority drug target population in Africa [52, 61]. Although 90% of schistosomiasis infected people live in Africa there have been no studies of PZQ population genetics in Africa [95, 96], highlighting the need for pharmacogenetic studies on the clinically relevant CYP variants in target populations [94]. For example, CYP2D6 is a metabolic route of PZQ; therefore, the presence of CYP2D6\*17, which is unique to populations of African origin [97], exemplifies the need for its impact on PZQ metabolism and overall treatment efficacy to be evaluated. Nleya and colleagues [84] have provided evidence CYP1A2 may be the metabolic route of PZQ to its main metabolite in Zimbabwean volunteers. As a recent study reported that Zimbabwean children in schistosomiasis-endemic areas exhibited decreased CYP1A2 activity, and, as PZQ is metabolised via this pathway, this could also have detrimental implications on bioavailability [45].

In order to translate molecular findings into drug metabolism and predict efficacy, there is a need to characterise the PZQ metabolites, their quantities and their effect. Wang and colleagues [28] have performed the most comprehensive evaluation of PZQ's metabolites, albeit in experimental models, which identified phase I and phase II metabolites using both in vivo and in vitro methods. Fifteen metabolites were structurally identified from urine and faeces of mice 24 hours after PZQ dosing. The in vitro incubations using human liver microsomes (HLM) and human recombinant enzymes confirmed the metabolic activity of CYP1A2, CYP2C9, 2C19, 2D6, and 3A4/5, with metabolic products identified via all of these pathways. However, the *in vivo* results from the experimental studies were not identical to the human *in* vitro HLM, with 3 dioxidised metabolites not detected in HLM. The Zimbabwean study, one of the few human studies, also characterised the metabolic pathway of PZQ to its main metabolite: 4-OH-PZQ [84]. The combination of PZQ with CYP3A4/5 inhibitor ketoconazole resulted in significant changes in the AUC of the main 4-OH-PZQ metabolites, with increases of 57% (cis-) and 67% (trans-) when co-dosed with ketoconazole. The overall increase in the AUC of both 4-OH-PZQ metabolites revealed that when the CYP3A4/5 pathway was inhibited, there was a greater exposure of the main metabolite in the circulation. This indicates that CYP3A4/5 is not the metabolic route of 4-OH-PZQ and suggests that other CYP pathways, primarily CYP1A2 or CYP2C19, are instead involved in the principal elimination of the active parent drug. So what metabolite is the CYP3A4/5 pathway producing? The study reported a novel metabolite, X-OH-PZQ, which was suppressed upon administration of ketoconazole with PZQ, and therefore appeared to be dependent on CYP3A4/5 for its formation. The X-OH-PZQ levels were reduced by approximately 57%, providing categorical evidence that this novel metabolite is produced via the CYP3A4/5 pathway. Other studies have discussed the use of CYP3A inhibitors to reduce the conversion of PZQ to its main metabolite (Fig 7[II]) [53]; however, based on the results from this study, it appears that the introduction of a CYP1A2/CYP2C19 inhibitor to PZQ may result in a greater exposure of active parent drug. If the 4-OH-PZQ metabolic pathway is inhibited, then a reduction in the elimination of PZQ could occur and result in higher circulating concentration of the parent drug enhancing therapeutic efficacy. The exact structure of X-OH-PZQ was, at this time, not yet fully determined; to achieve this, the exact nature of the hydroxylation biotransformation must be evaluated. The intricacies of metabolite identification is potentially one of the factors contributing to the limited number of results addressing this topic, as Schepmann and colleagues [89] demonstrated, dedicating an entire paper to elucidating the structure of one phase I metabolite: 8-OH-PZQ (Fig 7[I]). By obtaining further information on PZQ's metabolites via the detection and quantification of each metabolite, they could be compared between individuals for variability. As individual genetic variation in the CYP enzymes could affect PZQ metabolite concentration in systemic circulation, there is the potential to use the metabolite itself as genetic marker without the need for sequencing.

As schistosomiasis is coendemic with several other parasites and pathogens, affected populations can be subjected to drug coadministration, giving rise to DDIs. Malaria is one such coendemic parasite, and coadministering the antimalarials mefloquine and artemether with PZQ was investigated in experimental models, building on work from an *in vitro* experiment by Keiser and colleagues [98]. The coadministration of reduced doses of PZQ + mefloquine and PZQ + artemether showed enhanced pharmacological activity and efficacy over PZQ alone [73–75]. Other coendemic infections include intestinal helminths, and helminth control programs often coadminister PZQ and albendazole or mebendazole [99]. In humans, the combination of the anthelmintic albendazole + PZQ resulted in an increase in the plasma concentrations of the active enantiomers of both drugs [54]. Previous reports presented contradictory data regarding the DDIs between PZQ and albendazole, concluding that coadministration of PZQ + albendazole does not alter PZQ PK [100, 101]. However, Lima and colleagues [54] demonstrated a synergistic interaction upon this combination with the AUC of the (R)-PZQ increasing by 76%, which would mean an increase in the exposure of the active drug to the schistosomes. This interaction was determined to be due to the albendazole inhibiting CYP1A and CYP3A pathways, indicating the (R)-PZQ cannot be metabolised and remains in systemic circulation longer, leading to higher (R)-PZQ plasma concentrations. Nonetheless, even with the advantage of increased PZQ exposure, the increased plasma concentrations of both active drugs risks unknown adverse effects without further investigation [54]. This beneficial combination is not the case with chloroquine. This antimalarial drug, when coadministered with PZQ, decreased the AUC of PZQ by approximately 64% compared to PZQ alone [86] and may create too low a systemic concentration of PZQ to be lethal to schistosomes. This antagonistic effect on PZQ metabolism was deemed not to be due to CYP alterations, highlighting the need for further investigations into this DDI mechanism.

Other drugs affecting CYP activity have also been shown to synergistically affect PZQ plasma availability. For example, cimetidine more than doubled the plasma levels of PZQ during combined drug administration than PZQ alone. Cimetidine nonselectively inhibits CYP1A2, 2C9, 2D6, and 3A4 [102], all of which are metabolic pathways of PZQ. A nilutamide + PZQ combination was also found to be synergistic in rodent infections, significantly

increasing WBR by 67% compared with PZQ alone [76]. In humans, coadministering PZQ + ketoconazole can increase the duration of active PZQ exposure [84] by inhibiting the CYP3A4/5 pathway to reduce clearance of PZQ by 30% and enhance parasite exposure to PZQ [84]. The AUC value at 20 mg/kg PZQ when coadministered with ketoconazole was comparable with a study using 40 mg/kg PZQ alone [63]. Despite this, upon a linear extrapolation of the level of PZQ exposure from this combination, it predicted only a 37.5% reduction in PZQ dose. Therefore, regardless of this synergistic activity, there would not be a significant reduction of the current tablet size with this DDI and, so, would not be a better alternative than the development of a new formulation [84].

There is definite need for more studies of PZQ metabolism in humans, specifically in affected populations. The majority of studies reviewed here were experimental models of schistosomiasis, and findings from these cannot always be easily translated or extrapolated to human hosts, particularly the PZQ metabolite characterisation studies. Although animal studies are useful as models of potential DDIs, some enzymes that are orthologous to the CYPs are not representative of human responses [43]. A methodological limitation of this review was the measurement used. One of the main measures of exposure used to compare drug exposure in this review is the AUC; however, the absolute bioavailability is generally a more representative parameter. It is calculated using the AUC of the intravenous route and the AUC of the IV dose [103], but this data was not available for the majority of included PK studies. The quantity and type of data available in the original publications also limited the conclusions in this review, particularly when data were presented as the mean values postanalysis and not the raw data; therefore, no further statistical analysis was possible. Additionally, we could not identify clearly established drug-specific thresholds beyond which reductions in AUC might lead to alterations in drug effect. The use of the 20% threshold in change in AUC was a representative value to identify important changes in systemic bioavailability [87]. While the lack of defined thresholds may hinder the clinical interpretations in this review, there was no definitive method to determine the DDIs from the data available. Therefore, the DDI effects described here were based on efficacy of PZQ treatment stemming from exposure to the active drug and are not necessarily the confirmation of a mechanistic interaction. Also, the low number of published studies that were obtained for each topic using the search criteria highlighted the gap in the knowledge in PZQ metabolism during schistosomiasis treatment. When the search criteria used in this review was applied to tuberculosis, there was approximately eight times the number of search results (S12 Table). This increase in results could be due to the number of different drugs used in tuberculosis, in comparison to the one drug used for schistosomiasis; nonetheless, this shows the difference in the quantity of published research for schistosomiasis in comparison to a disease that can occur in every part of the world.

Overall, schistosomiasis control is predominantly reliant on a single drug, PZQ, for treating millions of people. Variable cure rates from the drug raise concerns about the possibility of the development of drug resistance amongst the parasites; therefore, there is a need to determine the sources of heterogeneity in cure rates and determine the relative contribution of host-related factors. Our review has shown that several such factors can result in variable levels of PZQ in systemic circulation that potentially contribute to these low cure rates. These include drug formulation (enantiomers) and brand, the health of the host' s liver, host age, coadministered drugs, and host genetics. There is need for more of these studies in affected human populations, especially in Africa, where host and parasite attributes are studied simultaneously, to fully understand the sources of heterogeneity in PZQ efficacy.

#### **Supporting information**

**S1 Fig. Flow chart guides by the PRISMA for a systematic review, including the records excluded based on the specific criteria.** PRISMA, Preferred Reporting Items for Systematic Reviews and Meta-Analysis. (TIFF)

' Fable Included studies obtai

S1 Table. Included studies obtained using the inclusion and exclusion criteria of this review, separated into topics of focus. (PDF)

**S2 Table. Comparison of different brands of PZQ on the PK parameters in a rodent model.** PK, pharmacokinetic; PZQ, praziquantel. (PDF)

**S3** Table. Comparison of different brands of PZQ on efficacy of treatment in an infected murine model. PZQ, praziquantel. (PDF)

**S4 Table.** Comparison of the WBR in an infected murine model when dosing racemic **PZQ**, or a single enantiomer. PZQ, praziquantel; WBR, worm burden reduction. (PDF)

**S5 Table.** Pharmacokinetic parameters of PZQ enantiomers and its major metabolites in **HNV** after administration of racemic PZQ, or the single PZQ enantiomer itself. HNV, healthy normal volunteers; PZQ, praziquantel. (PDF)

**S6** Table. Pharmacokinetic parameters in human models after administration of racemic **PZQ; includes P and HNV.** HNV, healthy normal volunteers; P, infected patients; PZQ, praziquantel.

(PDF)

**S7 Table.** Pharmacokinetic parameters of PZQ, PZQ enantiomers, and the major metabolite in HSAC, infected SAC, and infected PSAC after the administration of racemic PZQ. HSAC, healthy school age children; PSAC, preschool-aged children; PZQ, praziquantel; SAC, school-aged children.

(PDF)

**S8 Table. Efficacy parameters of racemic PZQ treatment on infected PSAC and SAC, and the effect of the different brands of PZQ available.** PSAC, preschool-aged children; PZQ, praziquantel; SAC, school-aged children. (PDF)

**S9** Table. The comparison of dosing racemic PZQ alone, or with drug "B" on PZQ's antischistosomal efficacy in rodent models. PZQ, praziquantel. (PDF)

**S10** Table. The pharmacokinetic parameters of PZQ, its enantiomers, and the major metabolite in HNV after the administration of racemic PZQ with drug "B" to evaluate potential DDIs. DDI, drug–drug interactions; HNV, healthy normal volunteers; PZQ, prazi-quantel.

(PDF)

S11 Table. The comparison of racemic PZQ alone, and with drug "B" in reducing hepatic granuloma diameter in murine models. PZQ, praziquantel.

(PDF)

S12 Table. Comparison of the results of this systematic review in comparison to tuberculosis.

(PDF)

#### Acknowledgments

Thank you to all the members of the Parasite Immuno-epidemiology Group at the University of Edinburgh for their useful comments during the development of this manuscript.

#### References

- Ross AGP, Bartley PB, Sleigh AC, Olds GR, Li Y, Williams GM, et al. Schistosomiasis. N Engl J Med. 2002; 346(16):1212–20. https://doi.org/10.1056/NEJMra012396 PMID: <u>11961151</u>
- Handzel T, Karanja DM, Addiss DG, Hightower AW, Rosen DH, Colley DG, et al. Geographic distribution of schistosomiasis and soil-transmitted helminths in Western Kenya: implications for anthelminthic mass treatment. Am J Trop Med Hyg. 2003; 69(3):318–23. PMID: 14628951
- Integrated preventive chemotherapy for neglected tropical diseases: estimation of the number of interventions required and delivered, 2009–2010. Wkly Epidemiol Rec. 2012; 87(2):17–27. PMID: 22242234
- Utzinger J, Raso G, Brooker S, De Savigny D, Tanner M, ØRnbjerg N, et al. Schistosomiasis and neglected tropical diseases: towards integrated and sustainable control and a word of caution. Parasitology. 2009; 136(13):1859–74. https://doi.org/10.1017/S0031182009991600 PMID: 19906318
- WHO. Sustaining the drive to overcome the global impact of neglected tropical diseases: second WHO report on neglected diseases. 2013, World Health Organization. http://www.who.int/iris/handle/ 10665/77950.
- Meyer T, Sekljic H, Fuchs S, Bothe H, Schollmeyer D, Miculka C. Taste, A New Incentive to Switch to (R)-Praziquantel in Schistosomiasis Treatment. PLoS Negl Trop Dis. 2009; 3(1):e357. https://doi.org/ 10.1371/journal.pntd.0000357 PMID: 19159015
- 7. Cioli D, Pica-Mattoccia L. Praziquantel. Parasitol Res. 2003; 90 Supp 1:S3-9.
- Fenwick A. Praziquantel: do we need another antischistosoma treatment? Future Med Chem. 2015; 7 (6):677–80. https://doi.org/10.4155/fmc.15.16 PMID: 25996058
- Thomas CM, Timson D. The mechanism of action of praziquantel: can new drugs exploit similar mechanisms? Curr Med Chem. 2018. <u>https://doi.org/10.2174/0929867325666180926145537</u> PMID: 30259811
- Doenhoff MJ, Cioli D, Utzinger J. Praziquantel: mechanisms of action, resistance and new derivatives for schistosomiasis. Curr Opin Infect Dis. 2008; 21(6):659–67. <u>https://doi.org/10.1097/QCO.</u> 0b013e328318978f PMID: 18978535
- Gönnert R, Andrews P. Praziquantel, a new broad-spectrum antischistosomal agent. Z F Parasitenkunde. 1977; 52:129–50.
- Kimani BW, Mbugua AK, Kihara JH, Ng'ang'a M, Njomo DW. Safety, efficacy and acceptability of praziquantel in the treatment of *Schistosoma haematobium* in pre-school children of Kwale County, Kenya. PLoS Negl Trop Dis. 2018; 12(10):e0006852. <u>https://doi.org/10.1371/journal.pntd.0006852</u> PMID: 30332403
- Zwang J, Olliaro PL. Clinical efficacy and tolerability of praziquantel for intestinal and urinary schistosomiasis-a meta-analysis of comparative and non-comparative clinical trials. PLoS Negl Trop Dis. 2014; 8(11):e3286. https://doi.org/10.1371/journal.pntd.0003286 PMID: 25412105
- 14. Midzi N, Sangweme D, Zinyowera S, Mapingure MP, Brouwer KC, Kumar N, et al. Efficacy and side effects of praziquantel treatment against *Schistosoma haematobium* infection among primary school children in Zimbabwe. Trans R Soc Trop Med Hyg. 2008; 102(8):759–66. <u>https://doi.org/10.1016/j.trstmh.2008.03.010</u> PMID: 18486169
- **15.** Barakat R, Morshedy HE. Efficacy of two praziquantel treatments among primary school children in an area of high *Schistosoma mansoni* endemicity, Nile Delta, Egypt. Parasitology. 2010; 138(4):440–6. https://doi.org/10.1017/S003118201000154X PMID: 21078223

- Tchuenté L-AT, Shaw DJ, Polla L, Cioli D, Vercruysse J. Efficacy of Praziquantel against Schistosoma haematobium infection in children. Am J Trop Med Hyg. 2004; 71(6):778–82. PMID: 15642971
- Alonso D, Munoz J, Gascon J, Valls ME, Corachan M. Failure of standard treatment with praziquantel in two returned travelers with *Schistosoma haematobium* infection. Am J Trop Med Hyg. 2006; 74 (2):342–4. PMID: 16474094
- Danso-Appiah A, De Vlas SJ. Interpreting low praziquantel cure rates of *Schistosoma mansoni* infections in Senegal. Trends Parasitol. 2002; 18(3):125–9. https://doi.org/10.1016/s1471-4922(01)02209-7 PMID: <u>11854090</u>
- Kabuyaya M, Chimbari MJ, Manyangadze T, Mukaratirwa S. Efficacy of praziquantel on Schistosoma haematobium and re-infection rates among school-going children in the Ndumo area of uMkhanyakude district, KwaZulu-Natal, South Africa. Infect Dis Poverty. 2017; 6(1):83. https://doi.org/10.1186/ s40249-017-0293-3 PMID: 28385154
- Sabah AA, Fletcher C, Webbe G, Doenhoff MJ. Schistosoma mansoni: chemotherapy of infections of different ages. Exp Parasitol. 1986; 61(3):294–303. <u>https://doi.org/10.1016/0014-4894(86)90184-0</u> PMID: 3086114
- Utzinger J, Keiser J, Shuhua X, Tanner M, Singer BH. Combination chemotherapy of schistosomiasis in laboratory studies and clinical trials. Antimicrob Agents Chemother. 2003; 47(5):1487–95. https:// doi.org/10.1128/aac.47.5.1487-1495.2003 PMID: 12709312
- 22. WHO. Report of the Scientific Working Group Meeting on Schistosomiasis 2005, World Health Organization. https://apps.who.int/iris/handle/10665/69482.
- Xiao SH, Yue WJ, Yang YQ, You JQ. Susceptibility of *Schistosoma japonicum* to different developmental stages to praziquantel. Chin Med J (Engl). 1987; 100(9):759–68. PMID: 3127152
- Vale N, Gouveia MJ, Rinaldi G, Brindley PJ, Gärtner F, Correia da Costa JM. Praziquantel for Schistosomiasis: Single-Drug Metabolism Revisited, Mode of Action, and Resistance. Antimicrob Agents Chemother. 2017; 61(5):e02582–16. https://doi.org/10.1128/AAC.02582-16 PMID: 28264841
- Cupit PM, Cunningham C. What is the mechanism of action of praziquantel and how might resistance strike? Future Med Chem. 2015; 7(6):701–5. https://doi.org/10.4155/fmc.15.11 PMID: 25996063
- 26. Cyprotex. Everything you need to know about ADME (2nd ed): Cyptrotex PLC; 2015.
- Hoffmann MF, Preissner SC, Nickel J, Dunkel M, Preissner R, Preissner S. The Transformer database: biotransformation of xenobiotics. Nucleic Acids Res. 2014; 42(Database issue):D1113–7. https://doi.org/10.1093/nar/gkt1246 PMID: 24334957
- Wang HN, Fang ZZ, Zheng Y, Zhou K, Hu CY, Krausz KW, et al. Metabolic profiling of praziquantel enantiomers. Biochem Pharmacol. 2014; 90(2):166–78. https://doi.org/10.1016/j.bcp.2014.05.001 PMID: 24821110
- Evans WE, Relling MV. Pharmacogenomics: Translating functional genomics into rational therapeutics. Science. 1999; 286(5439):487–91. <u>https://doi.org/10.1126/science.286.5439.487</u> PMID: 10521338
- Buzkova H, Pechandova K, Slanar O, Perlik F. Genetic polymorphism of cytochrome P450 and methods for its determination. Prague Med Rep. 2006; 107(4):383–93. PMID: 17402551
- Masimirembwa CM, Hasler JA. Genetic polymorphism of drug metabolising enzymes in African populations: implications for the use of neuroleptics and antidepressants. Brain Res Bull. 1997; 44(5):561–71. https://doi.org/10.1016/s0361-9230(97)00307-9 PMID: 9365799
- Linder MW, Prough RA, Valdes R. Pharmacogenetics: a laboratory tool for optimizing therapeutic efficiency. Clin Chem. 1997; 43(2):254–66. PMID: 9023127
- Blower P, de Wit R, Goodin S, Aapro M. Drug–drug interactions in oncology: Why are they important and can they be minimized? Crit Rev Oncol Hematol. 2005; 55(2):117–42. <u>https://doi.org/10.1016/j.</u> critrevonc.2005.03.007 PMID: 15890526
- Olliaro P, Delgado-Romero P, Keiser J. The little we know about the pharmacokinetics and pharmacodynamics of praziquantel (racemate and R-enantiomer). J Antimicrob Chemother. 2014; 69(4):863– 70. https://doi.org/10.1093/jac/dkt491 PMID: 24390933
- Cascorbi I. Drug interactions-principles, examples and clinical consequences. Dtsch Arztebl Int. 2012; 109(33–34):546–56. https://doi.org/10.3238/arztebl.2012.0546 PMID: 23152742
- **36.** Jeffrey D Lewis KAB. Cytochrome P450 Enzymes and Drug—Drug Interactions: An Update on the Superfamily. J Pharm Technol. 2006; 22(1):22–31.
- Fenwick A, Webster JP. Schistosomiasis: challenges for control, treatment and drug resistance. Curr Opin Infect Dis. 2006; 19(6):577–82. https://doi.org/10.1097/01.qco.0000247591.13671.6a PMID: 17075334

- Liberati A, Altman DG, Tetzlaff J, Mulrow C, Gøtzsche PC, Ioannidis JPA, et al. The PRISMA Statement for Reporting Systematic Reviews and Meta-Analyses of Studies That Evaluate Health Care Interventions: Explanation and Elaboration. PLoS Med. 2009; 6(7):e1000100. <u>https://doi.org/10.1371/journal.pmed.1000100 PMID: 19621070</u>
- 39. Tuntland T, Ethell B, Kosaka T, Blasco F, Zang RX, Jain M, et al. Implementation of pharmacokinetic and pharmacodynamic strategies in early research phases of drug discovery and development at Novartis Institute of Biomedical Research. Front Pharmacol. 2014; 5:174. <u>https://doi.org/10.3389/fphar.2014.00174 PMID: 25120485</u>
- Ortega VE, Meyers DA. Pharmacogenetics: implications of race and ethnicity on defining genetic profiles for personalized medicine. J Allergy Clin Immunol. 2014; 133(1):16–26. <u>https://doi.org/10.1016/j.jaci.2013.10.040</u> PMID: 24369795
- **41.** Sankar P, Cho MK, Mountain J. Race and ethnicity in genetic research. Am J Med Genet A 2007; 143A(9):961–70. https://doi.org/10.1002/ajmg.a.31575 PMID: 17345638
- Botros S, El-Lakkany N, el-Din SHS, Sabra AN, Ibrahim M. Comparative efficacy and bioavailability of different praziquantel brands. Exp Parasitol. 2011; 127(2):515–21. <u>https://doi.org/10.1016/j.exppara.</u> 2010.10.019 PMID: 21044626
- 43. Botros S, El-Din SHS, El-Lakkany NM, Sabra ANA, Ebeid FA. Drug-metabolizing enzymes and praziquantel bioavailability in mice harboring *Schistosoma mansoni* isolates of different drug susceptibilities. J Parasitol Res. 2006; 92(6):1344–9. https://doi.org/10.1645/GE-865R.1 PMID: 17304818
- Kokwaro GO, Taylor G. Effect of experimentally-induced hepatic cirrhosis on the pharmacokinetics of orally administered praziquantel in the rat. Eur J Drug Metab Pharmacokinet. 1990; 15(3):199–202. https://doi.org/10.1007/BF03190204 PMID: 2253649
- 45. Gotardo MA, Hyssa JT, Carvalho RS, De-Carvalho RR, Gueiros LS, Siqueira CM, et al. Modulation of expression and activity of cytochrome P450s and alteration of praziquantel kinetics during murine schistosomiasis. Mem Inst Oswaldo Cruz. 2011; 106(2):212–9. https://doi.org/10.1590/s0074-02762011000200016 PMID: 21537683
- 46. Zhang DL, Wang HN, Ji JB, Nie L, Sun DQ. A quantification method for determination of racemate praziquantel and R-enantiomer in rat plasma for comparison of their pharmacokinetics. J Chromatogr B Analyt Technol Biomed Life Sci 2017; 1048:64–9. <u>https://doi.org/10.1016/j.jchromb.2017.02.013</u> PMID: 28222334
- Lombardo FC, Perissutti B, Keiser J. Activity and pharmacokinetics of a praziquantel crystalline polymorph in the *Schistosoma mansoni* mouse model. Eur J Pharm Biopharm. 2019; 142:240–6. <a href="https://doi.org/10.1016/j.ejpb.2019.06.029">https://doi.org/10.1016/j.ejpb.2019.06.029</a> PMID: 31265895
- **48.** WHO. Monitoring Anthelmintic Efficacy for Soil Transmitted Helminths (STH) 2008, World Health Organization. https://www.who.int/neglected\_diseases/preventive\_chemotherapy/Chemotherapy\_anthelminthic\_drug\_WHO\_WB\_final.pdf.
- Meister I, Ingram-Sieber K, Cowan N, Todd M, Robertson MN, Meli C, et al. Activity of praziquantel enantiomers and main metabolites against *Schistosoma mansoni*. Antimicrob Agents Chemother. 2014; 58(9):5466–72. https://doi.org/10.1128/AAC.02741-14 PMID: 24982093
- Tanaka M, Ohmae H, Utsunomiya H, Nara T, Irie Y, Yasuraoka K. A comparison of the antischistosomal effect of levo- and dextro-praziquantel on *Schistosoma japonicum* and *S. mansoni* in mice. Am J Trop Med Hyg. 1989; 41(2):198–203. https://doi.org/10.4269/ajtmh.1989.41.198 PMID: 2505624
- Bagchus WM, Bezuidenhout D, Harrison-Moench E, Kourany-Lefoll E, Wolna P, Yalkinoglu O. Relative bioavailability of orally dispersible tablet formulations of levo- and racemic praziquantel: two phase I studies. Clin Transl Sci. 2019; 12(1):66–76. https://doi.org/10.1111/cts.12601 PMID: 30536632
- Bonate PL, Wang TL, Passier P, Bagchus W, Burt H, Lupfert C, et al. Extrapolation of praziquantel pharmacokinetics to a pediatric population: a cautionary tale. J Pharmacokinet Pharmacodyn. 2018; 45(5):747–62. https://doi.org/10.1007/s10928-018-9601-1 PMID: 30218416
- Lima RM, Ferreira MAD, Ponte TMD, Marques MP, Takayanagui OM, Garcia HH, et al. Enantioselective analysis of praziquantel and trans-4-hydroxypraziquantel in human plasma by chiral LC-MS/MS: Application to pharmacokinetics. J Chromatogr B Analyt Technol Biomed Life Sci 2009; 877 (27):3083–8. https://doi.org/10.1016/j.jchromb.2009.07.036 PMID: 19665946
- 54. Lima RM, Ferreira MAD, Carvalho T, Fernandes BJD, Takayanagui OM, Garcia HH, et al. Albendazole-praziquantel interaction in healthy volunteers: kinetic disposition, metabolism and enantioselectivity. Br J Clin Pharmacol. 2011; 71(4):528–35. <u>https://doi.org/10.1111/j.1365-2125.2010.03874.x</u> PMID: 21395645
- 55. el Guiniady MA, Eltouny MA, Abdelbary MA, Abdelfatah SA, Metwally A. Clinical and pharmacokinetic study of praziquantel in Egyptian schistosomiasis patients with and without liver cell failure. Am J Trop Med Hyg. 1994; 51(6):809–18. https://doi.org/10.4269/ajtmh.1994.51.809 PMID: 7810816

- 56. Kovac J, Meister I, Neodo A, Panic G, Coulibaly JT, Falcoz C, et al. Pharmacokinetics of Praziquantel in *Schistosoma mansoni-* and *Schistosoma haematobium*-Infected School- and Preschool-Aged Children. Antimicrob Agents Chemother. 2018; 62(8). https://doi.org/10.1128/AAC.02253-17 PMID: 29866859
- Ofori-Adjei D, Adjepon-Yamoah KK, Lindstrom B. Oral praziquantel kinetics in normal and Schistosoma haematobium-infected subjects. Ther Drug Monit. 1988; 10(1):45–9. PMID: 3131934
- 58. Bustinduy AL, Waterhouse D, de Sousa-Figueiredo JC, Roberts SA, Atuhaire A, Van Dam GJ, et al. Population Pharmacokinetics and Pharmacodynamics of Praziquantel in Ugandan Children with Intestinal Schistosomiasis: Higher Dosages Are Required for Maximal Efficacy. MBio. 2016; 7(4). <u>https://doi.org/10.1128/mBio.00227-16 PMID: 27507822</u>
- Jung H, Medina R, Castro N, Corona T, Sotelo J. Pharmacokinetic study of praziquantel administered alone and in combination with cimetidine in a single-day therapeutic regimen. Antimicrob Agents Chemother. 1997; 41(6):1256–9. https://doi.org/10.1128/AAC.41.6.1256 PMID: 9174180
- 60. Kovac J, Panic G, Neodo A, Meister I, Coulibaly JT, Schulz JD, et al. Evaluation of a novel micro-sampling device, Mitra (TM), in comparison to dried blood spots, for analysis of praziquantel in *Schisto-soma haematobium*-infected children in rural Cote d'Ivoire. J Pharm Biomed Anal. 2018; 151:339–46. https://doi.org/10.1016/j.jpba.2018.01.030 PMID: 29413983
- Leopold G, Ungethum W, Groll E, Diekmann HW, Nowak H, Wegner DH. Clinical pharmacology in normal volunteers of praziquantel, a new drug against schistosomes and cestodes. An example of a complex study covering both tolerance and pharmacokinetics. Eur J Clin Pharmacol. 1978; 14 (4):281–91. https://doi.org/10.1007/BF00560463 PMID: 729622
- Metwally A, Bennett J, Botros S, Ebeid F, el attar Gel D. Impact of drug dosage and brand on bioavailability and efficacy of praziquantel. Pharmacol Res. 1995; 31(1):53–9. https://doi.org/10.1016/1043-6618(95)80048-4 PMID: 7784306
- Mandour ME, el Turabi H, Homeida MM, el Sadig T, Ali HM, Bennett JL, et al. Pharmacokinetics of praziquantel in healthy volunteers and patients with schistosomiasis. Trans R Soc Trop Med Hyg. 1990; 84(3):389–93. https://doi.org/10.1016/0035-9203(90)90333-a PMID: 2124391
- Castro N, Medina R, Sotelo J, Jung H. Bioavailability of praziquantel increases with concomitant administration of food. Antimicrob Agents Chemother. 2000; 44(10):2903–4. https://doi.org/10.1128/ aac.44.10.2903-2904.2000 PMID: 10991886
- Zimmermann GR, Lehár J, Keith CT. Multi-target therapeutics: when the whole is greater than the sum of the parts. Drug Discov Today. 2007; 12(1):34–42. <u>https://doi.org/10.1016/j.drudis.2006.11.008</u> PMID: 17198971
- Li Y, Meng Q, Yang M, Liu D, Hou X, Tang L, et al. Current trends in drug metabolism and pharmacokinetics. Acta Pharm Sin B. 2019; 9(6):1113–44. <u>https://doi.org/10.1016/j.apsb.2019.10.001</u> PMID: 31867160
- Kang JS, Lee MH. Overview of therapeutic drug monitoring. Korean J Intern Med. 2009; 24(1):1–10. https://doi.org/10.3904/kjim.2009.24.1.1 PMID: 19270474
- Cokol M, Chua HN, Tasan M, Mutlu B, Weinstein ZB, Suzuki Y, et al. Systematic exploration of synergistic drug pairs. Mol Syst Biol. 2011; 7(1):544. https://doi.org/10.1038/msb.2011.71 PMID: 22068327
- Foucquier J, Guedj M. Analysis of drug combinations: current methodological landscape. Pharmacol Res Perspect. 2015; 3(3):e00149–e. https://doi.org/10.1002/prp2.149 PMID: 26171228
- 70. Aschenbrenner DS, Venable SJ. Drug therapy in nursing: Lippincott Williams & Wilkins; 2009.
- 71. Caesar LK, Cech NB. Synergy and antagonism in natural product extracts: when 1 + 1 does not equal 2. Nat Prod Rep. 2019; 36(6):869–88. https://doi.org/10.1039/c9np00011a PMID: 31187844
- 72. Roell KR, Reif DM, Motsinger-Reif AA. An Introduction to Terminology and Methodology of Chemical Synergy—Perspectives from Across Disciplines. Front Pharmacol. 2017; 8(158). https://doi.org/10. 3389/fphar.2017.00158 PMID: 28473769
- 73. EI-Lakkany NM, el-Din SHS, Sabra A, Hammam OA. Pharmacodynamics of mefloquine and praziquantel combination therapy in mice harbouring juvenile and adult *Schistosoma mansoni*. Mem Inst Oswaldo Cruz. 2011; 106(7):814–22. https://doi.org/10.1590/s0074-02762011000700006 PMID: 22124553
- 74. Utzinger J, Chollet J, You JQ, Mei JY, Tanner M, Xiao SH. Effect of combined treatment with praziquantel and artemether on *Schistosoma japonicum* and *Schistosoma mansoni* in experimentally infected animals. Acta Tropica. 2001; 80(1):9–18. https://doi.org/10.1016/s0001-706x(01)00138-3 PMID: 11495639
- 75. Botros S, Hammam OA, Mahmoud M, Bergquist R. Praziquantel efficacy in mice infected with PZQ non-susceptible *S. mansoni* isolate treated with artemether: Parasitological, biochemical and immuno-histochemical assessment. APMIS. 2010; 118(9):692–702. <u>https://doi.org/10.1111/j.1600-0463.2010</u>. 02645.x PMID: 20718722

- 76. Keiser J, Vargas M, Vennerstrom JL. Activity of antiandrogens against juvenile and adult Schistosoma mansoni in mice. J Antimicrob Chemother. 2010; 65(9):1991–5. <u>https://doi.org/10.1093/jac/dkq233</u> PMID: 20576641
- 77. El-Lakkany N, Nosseir M. Pharmacodynamics of pentoxifylline and/or praziguantel in murine schistosomiasis mansoni. APMIS. 2007; 115(3):184–94. <u>https://doi.org/10.1111/j.1600-0463.2007.apm\_501.x PMID: 17367463</u>
- 78. Botros S, Mahmoud M, Hammam O, Salah F, Zidek Z, Masek K. Effect of adamantylamide dipeptide as adjuvant therapy to praziquantel in mice infected with different *S. mansoni* isolates. APMIS. 2006; 114(7–8):480–91. https://doi.org/10.1111/j.1600-0463.2006.apm\_380.x PMID: 16907853
- **79.** Abla N, Keiser J, Vargas M, Reimers N, Haas H, Spangenberg T. Evaluation of the pharmacokineticpharmacodynamic relationship of praziquantel in the *schistosoma mansoni* mouse model: Possible clinical implications. Trop Med Int Health. 2017; 22 (Supplement 1):136–7.
- Hussein A, Rashed S, El Hayawan I, El-Sayed R, Ali H. Evaluation of the Anti-schistosomal Effects of Turmeric (Curcuma longa) Versus Praziquantel in *Schistosoma mansoni* Infected Mice. Iran J Parasitol. 2017; 12(4):587–96. PMID: 29317884
- Masimirembwa CM, Naik YS, Hasler JA. Effects of phenobarbital and 3-methylcholanthrene pretreatment on the pharmacokinetics of praziquantel in rats. Eur J Drug Metab Pharmacokinet. 1993; 18 (3):261–4. https://doi.org/10.1007/BF03188806 PMID: 8149944
- Araujo N, De Mattos ACA, Coelho PMZ, Katz N. Association of oxamniquine, praziquantel, and clonazepam in experimental *Schistosomiasis mansoni*. Mem Inst Oswaldo Cruz. 2008; 103(8):781–5. https://doi.org/10.1590/s0074-02762008000800007 PMID: 19148417
- Ebeid FA. Efficacy of a combination of Metrifonate and praziquantel in the treatment of Schistosomamansoni infection in albino mice. Arzneimittelforschung. 1995; 45-1(4):519–21.
- Nleya L, Thelingwani R, Li XQ, Cavallin E, Isin E, Nhachi C, et al. The effect of ketoconazole on praziquantel pharmacokinetics and the role of CYP3A4 in the formation of X-OH-praziquantel and not 4-OH-praziquantel. Eur J Clin Pharmacol. 2019; 75(8):1077–87. <u>https://doi.org/10.1007/s00228-019-02663-8 PMID: 31089768</u>
- Metwally A, Bennett JL, Botros S, Ebeid F. Effect of cimetidine, bicarbonate and glucose on the bioavailability of different formulations of praziguantel. Arzneimittelforschung. 1995; 45-1(4):516–8.
- Masimirembwa CM, Naik YS, Hasler JA. The effect of chloroquine on the pharmacokinetics and metabolism of praziquantel in rats and in humans. Biopharm Drug Dispos. 1994; 15(1):33–43. https:// doi.org/10.1002/bdd.2510150103 PMID: 8161714
- Mills E, Montori VM, Wu P, Gallicano K, Clarke M, Guyatt G. Interaction of St John's wort with conventional drugs: systematic review of clinical trials. BMJ. 2004; 329(7456):27–30. https://doi.org/10.1136/ bmj.329.7456.27 PMID: 15231618
- Godawska-Matysik A, Kiec-Kononowicz K. Biotransformation of praziquantel by human cytochrome p450 3A4 (CYP 3A4). Acta Pol Pharm. 2006; 63(5):381–5. PMID: 17357589
- Schepmann D, Blaschke G. Isolation and identification of 8-hydroxypraziquantel as a metabolite of the antischistosomal drug praziquantel. J Pharm Biomed Anal. 2001; 26(5–6):791–9. https://doi.org/10. 1016/s0731-7085(01)00470-8 PMID: 11600290
- Ouellette RJ, Rawn JD. Alkanes and Cycloalkanes: Structures and Reactions. In: Ouellette RJ, Rawn JD, editors. Organic Chemistry. Boston: Elsevier; 2014. p. 111–61.
- **91.** Consortium PP. The pediatric formulation. Pediatric Praziquantel Consortium. 2020. https://www.pediatricpraziquantelconsortium.org/pediatric-formulation.
- 92. Coulibaly JT, Panic G, Silué KD, Kovač J, Hattendorf J, Keiser J. Efficacy and safety of praziquantel in preschool-aged and school-aged children infected with *Schistosoma mansoni*: a randomised controlled, parallel-group, dose-ranging, phase 2 trial. Lancet Glob Health. 2017; 5(7):e688–e98. <a href="https://doi.org/10.1016/S2214-109X(17)30187-0">https://doi.org/10.1016/S2214-109X(17)30187-0</a> PMID: 28619227
- Khojasteh S.C. WH, Hop C.E.C.A. Drug Metabolism and Pharmacokinetics Quick Guide: Oral Absorption. New York, NY: Springer; 2011.
- Zhou Y, Ingelman-Sundberg M, Lauschke VM. Worldwide Distribution of Cytochrome P450 Alleles: A Meta-analysis of Population-scale Sequencing Projects. Clin Pharmacol Ther. 2017; 102(4):688–700. https://doi.org/10.1002/cpt.690 PMID: 28378927
- 95. WHO. Schistosomiasis: World Health Organization; 2018. <u>http://www.who.int/en/news-room/fact-sheets/detail/schistosomiasis</u>.
- **96.** Mduluza T, Mutapi F. Putting the treatment of paediatric schistosomiasis into context. Infect Dis Poverty. 2017; 6(1):85. https://doi.org/10.1186/s40249-017-0300-8 PMID: 28388940

- Masimirembwa C. The Metabolism of Antiparasitic Drugs and Pharmacogenetics in African Populations. In: Gurib-Fakim A. EJ, editor. Molecular Mechanisms to Clinical Applications. Chemistry for Sustainable Development in Africa: Springer, Berlin, Heidelberg; 2013. p. 27.
- Keiser J, Manneck T, Vargas M. Interactions of mefloquine with praziquantel in the schistosoma mansoni mouse model and in vitro. J Antimicrob Chemother. 2011; 66(8):1791–7. https://doi.org/10.1093/ jac/dkr178 PMID: 21602552
- Albonico M, Levecke B, LoVerde PT, Montresor A, Prichard R, Vercruysse J, et al. Monitoring the efficacy of drugs for neglected tropical diseases controlled by preventive chemotherapy. J Glob Antimicrob Resist. 2015; 3(4):229–36. https://doi.org/10.1016/j.jgar.2015.08.004 PMID: 27842865
- Homeida M, Leahy W, Copeland S, Ali MM, Harron DW. Pharmacokinetic interaction between praziquantel and albendazole in Sudanese men. Ann Trop Med Parasitol. 1994; 88(5):551–9. <u>https://doi.org/10.1080/00034983.1994.11812903</u> PMID: 7979646
- Pengsaa K, Na-bangchang K, Limkittikul K, Kabkaew K, Lapphra K, Sirivichayakul C, et al. Pharmacokinetic investigation of albendazole and praziguantel in Thai children infected with Giardia intestinalis. Ann Trop Med Parasitol. 2004; 98:349–57. https://doi.org/10.1179/000349804225003398 PMID: 15228716
- 102. Rodrigues AD. Drug–Drug interactions. Eur J Pharm Biopharm. 2009; 71:277–94.
- Abo-EL-Sooud K. Absolute and Relative Bioavailability. In: Drug Discovery and Evaluation: Methods in Clinical Pharmacology: Springer; 2018. 1–7 p.